The Role of Quality of Life: Economic Development and Employee Recruitment and Retention in Prince George, BC

Patrick Barrette

B. Comm., University of Northern British Columbia, 2001

Project Submitted In Partial Fulfillment Of

The Requirements For The Degree Of

Master Of Business Administration

The University Of Northern British Columbia

April 2007

© Patrick Barrette, 2007

UNIVERSITY OF NORTHE FAN BRITISH COLUMBIA LIBRARY Prince George, B.C. **Executive Summary**

Acknowledgments

Section I Introduction

Section II Background and Case Study of Prince George, BC

Economic Characteristics of Prince George Population and Employment in Prince George

Economic development strategies

Section III Literature Review

Quality of Life Defined

Migration and Quality of Life

"Creative Class" by Richard Florida

Recruitment and Retention Issues in Prince George

Section IV Studies of Business Location Decisions

Quality of Life and Location Decisions

Impact of Quality of Life on Recruitment and Retention Quality of Life and Economic Development Strategy

Section V Research Methodology

Key Informant Interviews

Interview Results and Findings

Section VI Discussion of key findings against expectations from the literatures

Section VII Conclusions

Limitations and Further Research/Key Results

References

Appendix A Interview Questions for Key Informants

Appendix B Interview Consent Form

Executive Summary

Traditional methods of economic development and employee recruitment and retention in Prince George has met with varying success. There is mounting concern regarding the challenges encountered in recruitment of key personnel, skilled employees and specialists across various industries. This challenge has significant impact on employers of all sizes, and affects the local economy in addition to the social aspects of the community. Quality of life (QOL) factors in the community is becoming an important factor in employee recruitment and retention, and relevant literature suggests that quality of life is becoming an increasingly important consideration in modern business location decisions.

The basis of study in this paper has two parts: 1) a review of the literatures on the relationship of QOL in recruitment and retention of employees and economic development practices, and 2) a research on the impact of QOL on employers in the Prince George region. The study includes interviews with 10 respondents from various firms located in the Prince George area. The literature review suggests that QOL is becoming an increasingly important determinant in employee recruitment and retention, while there is limited evidence of its effectiveness as part of an economic development strategy. The research on the importance of QOL has been difficult to amass, due to the nature of its definition and measures used to provide a basis for policy and strategies. However, it is important for local decision makers to understand the importance QOL factors has on employee recruitment, and how it influences firms of various sizes, industries, and corporate function, in addition to the economic and social health of the community.

Acknowledgements

I would like to express my appreciation for the guidance of my supervisor, Dr. Greg Halseth, during this project. I would also like to thank my grandmother, May Barrette, for her dedication and instilled values to pursue higher education. Lastly, I thank by wife, Nicolle, for her support and encouragement throughout the entire MBA program; and my daughter, Myra, for giving me the motivation to complete this project.

Section I Introduction

The purpose of this study is to identify how quality of life (QOL) factors relate to the employee recruitment and retention challenges in Prince George. By determining this relationship, opportunity arises to consider the aspects of QOL, and how the region can initiate improvements to help the area attract new residents and for firms to attract personnel. Through this relationship I had anecdotal considerations in the idea that quality of life can be used as an economic development strategy, by creating an environment that is attractive to broad groups of people who seek to live in Prince George. The result is the creation of an attractive region for enterprises to locate and benefit from the abundant supply of potential employees. These ideas will be studied in literature reviews and research interviews to determine the relevance and impact of QOL.

QOL is difficult to define due to the vast number of factors and attributes that contribute to this concept. Most studies and researchers have concentrated on a few selected contributors of QOL such as education, healthcare, recreation, cost and standard of living, cost of housing, weather factors, remoteness and commuting, and a variety of recreational and cultural services and activities (Hack, 1984; Bramlage, 1988; Boyle, 1998, Festervand *et al.*, 1998). This list is not intended to represent a comprehensive list of contributors of QOL, but rather a sample of the items considered in this study. The impact of local employee shortages has both social and economic consequences to the residents and employers in the region. Examples of such shortages have been documented. In 2003, the loss of a local pathologist resulted in delays in RCMP investigations as autopsies had to be processed outside of community (Way 2005). Even with a targeted campaign, over a two

year period from 2000 to 2002, the Northern Health Authority (NHA) achieved a net gain of only seven doctors to Prince George. The NHA recruited 32 doctors and specialists during the period, but 25 left the community (Paulsen, 2003). As Way (2005:1) notes, "While shortages of health professionals received the most attention, a range of other professional positions remained difficult to fill, including university professors, lawyers and city planners".

While Prince George is characterized as a resource based region, there are factors that contribute to a broader economic development with the larger population, such as government and regional services providing additional amenities, like the regional hospital and university, which are not found in smaller resource communities. Yet, persistent longterm challenges in recruiting and retaining professionals are primary concerns of human resource managers in organizations, as well as concerns for the community as a whole. Local governments are engaged in a variety of initiatives and activities to promote economic development in an attempt to reduce the traditional reliance on one resource, thus ensuring stable growth in the future. These activities are designed to improve the region's overall QOL in traditional provisions of infrastructures and basic recreational facilities. Local municipal government plans and economic development strategies are based on the implicit view of the firms as a cost-minimizing agent. In an attempt to foster a 'business friendly' environment by using a number of approaches offering favorable tax incentives, interest-free development bonds, favorable lease arrangement, and other cost reducing incentives, the city hopes to reduce the cost of business and successfully lure investment away from other areas. Despite the continued practice and popularity of fiscal incentive,

there is empirical research that indicates that it is relatively ineffective in altering business location decisions (Blair and Premus, 1987; and Morgan, 1964). There is also the suggestion that incentives drain scarce fiscal resources that could be spent on other development strategies, such as upgrading infrastructures and developing local amenities as general improvements to the QOL (Salvenson and Renski, 2002).

There is a growing body of research that suggests QOL is becoming an increasingly important consideration for business location decisions. These views also indicate that QOL, among other factors, influence business location decisions and is considered by firms as important in attracting and retaining a high-quality work force. This view supports my ideas of QOL as a unifying paradigm that satisfies the need for attracting business investment, while simultaneously working to improve the lives of existing and future residents. Despite these views, there is limited empirical evidence of the effectiveness of an economic development strategy based on QOL, while there is strong support to indicate that QOL is an important factor in attracting and retaining employees in communities and regions.

Scope of the Paper

The purpose of this paper is not to debate the merits of current economic development or employee recruitment and retention strategies. By looking at the current practices of employers and economic development agencies in the region, we are able to examine the specific challenges being faced, and identify some contributing factors to the problems while providing alternative views and suggestions for addressing these challenges. I trust

that this paper will provide relevant knowledge and serve as a useful foundation for future study. This study should be of interest to employers, policy makers, and agencies in the region involved in local and regional economic development.

This paper is organized into six sections and a conclusion. Section II provides background on Prince George, including a discussion of economic and demographic characteristics along with some description of the local economic development strategy. Section III provides further insight into the components of QOL, the impact of QOL migration and its relationship with the emerging group knowledge workers referred to as the 'creative class.' Recruitment and retention issues in Prince George will also be reviewed and discussed. Section IV provides the body of knowledge that has been surveyed in helping to understand the concepts related to business location decisions, the impact of QOL on location decisions, recruitment and retention, and economic development strategies. Business location theory and the traditional site selection process are well reviewed by policy makers, and as a result this section is brief. Section V reviews the research methodology used in this paper and provides a detailed account of the influence of QOL on employee recruitment and retention as described by research respondents. A summary of responses in the interviews will be assessed, related to QOL with variances according to industry, corporate function, firm size, and specific organization characteristics. A summary of the results and findings are discussed in this section. Section VI links the key findings with discussions in the literature as it relates to QOL issues of and on recruitment and retention in Prince George. Finally, section VII of the paper concludes by summarizing the major

findings of the study, offering recommendations for application of the research results and areas for future study, and some limitations encountered in the research paper.

Section II

Economic Characteristics of Prince George Population & Employment characteristics Economic Development Strategies

This introductory section will review and discuss briefly the economic background of Prince George, the trends in population and employment in the region, and the economic development strategy used to address the issues of economic diversification and unemployment in the region.

Prince George is considered 'BC's Northern Capital' based on the central geographic location, size, and level of services along with the economic region's level of transportation infrastructure. The natural resources and the lumber industry have been essential in the city's economic development. Prince George is also a regional hub in transportation, connecting the major Vancouver metropolis through the regional center to the province's northern portion via railway and highway; in addition, providing access to the northwest coast of British Columbia through to Alberta in the east. The lumber and pulp and paper industries have had significant effects on regional growth and the local economy.

Prince George has developed as a resource-based town, highly dependent on the forest industry, with a high percentage of industrial developments around the city in the primary industry sector. This has led to the local economic cycles of rise and fall depending on the global markets for local forest products (Mullins, 1967, Hayter, 1976, Marchak, 1989).

With the growth of the primary resource sector, supporting services and commercial sector employment followed and has resulted in spin-off growth in supporting industries generally termed 'multiplier effect' from the basic resource industry (Halseth, 1991). This growth resulted in rapid increases in population from the periods of mid 1960s to the early 1980s. There have been contributing factors in population growth and economic development that are important events in the growth and development of Prince George. These include the changing forest industry, the development of the College of New Caledonia, and the opening of the University of Northern British Columbia.

Prince George Population

Prince George has primarily been a resource base industry town and as result been very susceptible to economic market fluctuations. It was reported by Halseth (1991) that during the early 1980s, an economic recession in the forest industry had a noticeable impact on the Prince George population. Economic development in the community slowed considerably and the city's population remained virtually unchanged between 1981 and 1986. By the early 1990s the forest industry had recovered from the recession and the population stagnation of the early 1980s was ended. There has been recent information released by Stats Canada and BC Stats that suggest different population figures for Prince George in 2006. As reported by Mark Nielsen of the Prince George Citizen,

While StatsCan census figures released Tuesday showed Prince George's population at 70,981 as of 2006, B.C. Stats' says 77,343 were living in the city, according to figures released in December. Moreover, while the census says the city's population is two per cent smaller than in 2001, B.C. Stats says it grew by 2.3 per cent.

Also reported by Dave Paulson of the Prince George Citizen,
Instead of a modest population increase that might have been expected, Prince
George showed a drop of two per cent since 2001, or about 1,400 residents, to
a shade under 71,000. The city's "metropolitan area," which includes
surrounding areas outside city limits, also fell by two per cent to 83,225.

This decrease in the city population was commented by Greg Halseth, "the decrease is in keeping with a pattern for most of northern B.C. and the advent of a more efficient resource sector that needs fewer workers". These comments help to summarize the trends in the population changes and the impact of the resource economy on the population of the city.

Demographic Characteristics of Prince George

Demographic information provided by Statistics Canada illustrate the characteristics of a community of young families. This feature is common in many resource-based communities that attract young families and workers to higher paying jobs. The general demographic trend in Prince George is a maturing and aging population; a trend in line with the rest of the province based on the aging 'Baby Boomer' generation A significant difference between the Prince George population and the BC average reveals that the percentage of people over age 65 represents just 7.6% of the population, compared to 13.6% for provincial average. A discussion with local economic development agencies attributed this variance to the lack of senior housing and home support, in addition to the common practice of retired professionals relocating to warmer climates in regions with higher levels of amenities and services for seniors. A growing need that has, and will, develop in the future as the population of Prince George ages increasing the demand for health services and amenities. Prince George also has a higher percentage than the provincial average age of people between the ages of 0 and 45. The university and college

has also contributed to the retention and increase of people between the ages of 18 and 24 as a result of students staying and coming to the city to attend school. In the past, this demographic group would leave the city to attend post secondary education. Population and demographic changes in Prince George has been shaped by economic restructuring and investments in its recent development.

Table 1.1
Population for Prince George & Age Characteristics including BC: 2001

Population Estimates**		Age Distribution	Prince George	
Year	Prince George	2001 Census**	Male	Female
2001	75,568	All Ages	36,240	36,165
2002	75,288	0-14	8,015	7,420
2003	76,641	15-24	5,675	5,560
2004	77,826	25-44	11,230	11,760
2005	77,148	45-64	8,750	8,470
		65+	2,590	2,955

Source: BC STATS

Source: Statistics Canada

Employment in Prince George

This section will examine the issues of employment and labour industries in Prince George. The information from Statistics Canada shows that 13.6% of the population is directly employed in the Logging and Forest Products industry in comparison to the provincial percentage of 4.7% (Statistics Canada, 2006). Despite continued efforts in local economic diversification, the forestry sector continues to be a major economic sector in Prince George. Canfor is the largest employer in the Prince George region with over 3056 employees. There are questions and concerns regarding the continuing role of the forest sector, with the mountain pine beetle infestation and its potential future economic impact. Other major employers include the Prince George Regional Hospital, Northern Health Authority (NHA), School District 57, the Government of BC, and the City of Prince George. One major sector with strong economic and social impacts is the education sector

including the University of Northern British Columbia (UNBC) and the College of New Caledonia (CNC). Combined, the university and college employ over 1400 people. Not only has the university and college been major employers, they have contributed significantly to the local economy through enhancement of faculty, staff, and institutional spending combined with earnings of alumni in the local economy.

Despite the growth in industries beyond forestry and lumber, the unemployment rate for the Prince George region has consistently been above the provincial average over the last 5 years. Prince George also features a lower percentage of workers compared to the provincial average in the industries related to arts, entertainment and recreation, scientific and technical services, and information and cultural industries. Prince George has previously experienced considerably higher than provincial average unemployment rates due to economic downturns and declines in the lumber industry. During 1998, the unemployment rate in Prince George reached 16.6%, double the provincial and national average of 8.2% and 8.0% respectively. This trend continued through to current statistical data. The most recent information from Statistics Canada reveals a Prince George unemployment rate of 5.8%, higher than the provincial average of 4.0% (seasonally adjusted) as of March 9, 2007 (Statistic Canada, 2007).

Economic Development Strategies of Prince George

In an effort to address high unemployment rate 2003, the City of Prince George and its local economic development agency, Initiatives Prince George (IPG), launched a campaign to attract the Synovate call center to invest in the region. The economic development

strategy was a 'quick fix' (Initiative Prince George, 2004) to local economic challenges. Initiatives Prince George was created in 2000 and is a municipally-owned corporation mandated to undertake programs and projects designed to grow and diversify the local economy. The primary economic development strategy adopted by Initiatives Prince George (2005) is to promote core services comprised of: Business, Export, Tourism, Events, Film, and City Center Ventures. Investments and activities have been focused on attracting industries and employers to address the unemployment challenges and diversify the local economy. Initiatives Prince George has experienced success in attracting call centers and big box retail to the city, helping to meet the goals of employment and economic diversification. While this is a success, the question remains: How can Prince George attract employers in professional and technology based industries? Such an attraction will further address the issues of economic diversification and employment in skilled and knowledge based industries where educated professionals are hired This goal is a better economic development option when compared to industries dependent on employment in low wage, casual, or part time positions targeted towards students or low skilled employees. One major economic success for Prince George has been the development of the university and college. The economic impact of CNC and UNBC in Prince George was \$721.4 million in 2004/05. This figure includes the combination of faculty and staff spending, student spending, institutional spending, visitor spending and alumni earnings (Prince George's Education Economy, 2006)

Currently there are numerous initiatives underway through Initiatives Prince George which presents opportunities for the region. A challenge faced by Prince George and its region is attracting and retaining skilled and trained professionals. There are efforts by the City of Prince George and IPG to market the community to potential residents and employers the city amenities, its resources, and available infrastructures that include transportation and communication systems as well as overall QOL. In reviewing the profile of Prince George, the challenges related to its economic condition and industries, and the goals of improving employment and economic development through diversification; this paper attempts to raise questions and potentially provide some alternatives and recommendations.

Many firms are adapting to the new economic environment as a result of a number of factors: the strong economic condition of BC, increased use of technology in industries, a high demand and competition for skilled labour, and industry shifts to knowledge workers. These factors have also contributed to the difficulties faced by firms in employee recruitment and retention that will be examined further in the study, and in the responses of the interviews. We will look at the public service and QOL factors in Prince George and examine the issues of recruitment and retention challenges. The focus of the paper is to help answer the following questions:

Can infrastructure investments supporting improved QOL in Prince George be used as an economic development strategy?

Can this strategy help to address the challenges related to Recruitment and Retention in Prince George?

This section reviewed the economic and demographic aspects of Prince George as a resource based community, the economic impact of the forest industry in the region, and the challenges to expand and diversify outside of this industry. The employment

characteristics have been heavily influenced by the forest industry in the past, and the trends in an aging population present potential challenges that will need to be addressed in the replacement of a retiring 'baby boomers' workforce. The College of New Caledonia and the University of Northern British Columbia have become major employers and have significantly contributed to its economic health by helping to diversify the regional industries and attract students and residents to the area. The literature review on the economic development and migration theories related to QOL and research interviews will assist in identify some of the challenges employers encounter in the recruitment and retention process.

Sections III
Defining Quality of Life
Migration and Quality of Life
"Rise of the Creative Class" and Implications on Economic Development
Recruitment and Retention Issues in Prince George, BC

This section will review and identify some common patterns that contribute to QOL, the existing factors in Prince George that determine QOL, and review of those studies completed over a 30 year period related to employee migration and QOL. The recent publications and concepts of Richard Florida in the 'Creative Class' will be reviewed to further identify the potential impact and opportunities for Prince George. Employee recruitment and retention issues in Prince George will be discussed along with a review of research on this subject completed by Laura Way in 2005.

A general term used to describe (QOL) is the social and economic well being of the population. QOL is very difficult to define due to the vast number of factors and attributes

that contribute to this concept. Most studies and researchers have concentrated on a few selected contributors of QOL. Festervand *et al.* (1998:20) noted: "Despite the increasing importance of qualify of life factors, there is little, if any, agreement among industrial site location professionals as to what this nebulous concept really is." The phrase is commonly used to describe an array and combination of qualities that contribute to a population's well being, such as education, healthcare, recreation, cost and standard of living, cost of housing, weather factors, remoteness and commuting, social and economic environment, childcare, pollution and crime, environment, and a variety of recreational and cultural services and activities. (Hack, 1984; Bramlage, 1988; Boyle, 1998, Festervand *et al.*, 1998). These factors were included in the definition used to describe QOL in the interview questions. However, it was not intended to represent a comprehensive list of contributors of QOL, but rather as a sample of the items considered. Education is expected to be one of the most valued QOL attractions (Brown, 1987; Rooney, 1986).

Prince George benefits from the recreational aspects that are associated with the region's natural and geographical characteristics. The outdoor enthusiast will enjoy the activities associated with lakes and mountainous terrain, such as downhill skiing, hiking, mountain biking, cross country skiing, snowmobiling, fishing, camping, etc.. Other QOL factors that Prince George offers are the relatively lower cost of housing compared to major regions and cities such as Kamloops, Kelowna, Vernon, Vancouver, and Victoria. The relatively smaller urban sprawl dictates much shorter commutes to and from work. Cost of living, housing, culture, restaurants, cinemas, and local live theatre entertainment have all been

shown to be important QOL factors that relate to the recruitment of key personnel (Browning, 1980).

The cost of housing, cost of living and short commute are key QOL benefits that are being recognized and marketed to potential employees from outside of Prince George by employers in the region. It is also emphasized that Prince George is 'a good place to raise a family' for some of the reasons of low housing cost, short commute, and low crime relative to major cities in BC. It has been noted that before QOL can be used by economic development officials to market their community to potential businesses, they must identify the elements that constitute QOL and the relative importance of each to their potential targeted business (Festervand et al., 1998). An area of QOL that is improving in Prince George is the increase of cultural and social diversity. Prince George has updated and expanded the Art Gallery, the Prince George Symphony, and the Multiplex, which hosts a number of concerts, sporting, and entertainment events. There has also been a parallel increase in the number of ethnic restaurants in the region, providing further diversity of amenities. Myers (1987) has claimed that items normally considered critical, such as entertainment, culture, and restaurants, are not as influential from a relative perspective when compared to education and cost of living.

QOL aspects in Prince George that are considered negative include the air quality, which receives regular media attention, remoteness or distance from other major city centers, the winter climate, city appearance (particularly the downtown core), limited entertainment and culture, and lack of a cohesive community image/spirit. The Prince George air quality is

the most significant 'push factor' found by Laura Way (2005) in her study of recruitment and retention challenges in Prince George. It is considered the most significant QOL factor for many existing and potential residence of Prince George. It was noted by Bamberger & Parham (1984) that some cities are actively pursuing economic development policies that target what is referred to as 'amenity infrastructure' investments that are entertainment The local climate is considered important to local labour pool attraction and Graves (1980) would suggest it to have a direct impact on recruitment and retention challenges experience in Prince George. Image and community spirit are elements of the local environment that includes the public perceptions of the appearances and image of the city and regions. Prince George has continued to be challenged with downtown business revitalization, and creating a positive image for the city core. In fact, cities often make extensive investments as a matter of public policy for the protection or development of local image (Bamberger and Parham, 1984). The redevelopment of the downtown core will make significant improvements to the perceptions of Prince George and its image as a destination city. This vital economic decision could improve the downtown revitalization initiative efforts.

Migration and Quality of Life

This section will briefly review the work completed by Alex Michalos in 1996, the relationships between migration, or residential mobility, and the QOL in Canada and the United States. It has been determined that migration is more prevalent among professional workers than managers and proprietors, and more prevalent among the latter than among labourers (Bogue, 1969). One would expect that migration increases with the education

level, since education is related to occupation type (Long, 1973). It was also found that the larger the family, the less likely it is for the family to move; whereas unmarried, divorced, or widowed persons are more likely to relocate then married people (Zuiches, 1980). Research suggests that the primary reason for migration is to seek an economic advantage, but people can be motivated to move to another region for reasons other than economics. Fuguitt (1985, p. 270) has written:

Economists continue to argue the 'chicken-egg' question of whether people follow jobs or jobs follow people...This research has provided evidence that both processes are at work and so opens the door to suggestions that amenity-based, or non-economically motivated, migration may result in increased economic opportunities.

This statement is summarized by Michalos in that while people may be motivated to move for some reason that is not economic, they also find some economic advantage in moving. This is referred to as the Tiebout (1956) hypothesis in economic literature. It suggests that individuals or households choose to live in communities that provide the maximum benefit of public goods and services at the minimum cost, assuming all other factors are roughly equal.

A summary of studies reviewed provide substantial evidence that people tend to move in order to improve the quality of their lives in a variety of specific respects, and they continue to move until they achieve majority of their goals (Michalos, 1996) The most frequently reported motive for relocating was job-related, including searching for jobs and for locations closer to a jobsite. The presence of family, friends, and people of the same ethnic or racial group is considered attractions at the receiving site. Other attractions include mild climate, minimum pollution, low crime rates, good health services, and good

educational facilities. It was found that job-related and house-related responses were cited as the top two reasons for both planning to move, and as reasons for picking a particular area. It was also found that compared to the average citizen of any country, migrants are typically younger, better educated, employed in secondary industries or professions, unmarried or recently married with comparatively fewer children, and are renters rather then home owners. There is also evidence from economic literature to indicate that people migrate to areas with relatively good provision of public goods and services, such as social security, educational facilities, and clean, safe, healthy environments. However, it was noted by Michalos that because areas with expanding public goods and services are typically also those with expanding economic opportunities. The fact that people migrate to such areas is not conclusive evidence that they are migrating primarily in the interests of obtaining such goods and services. This idea will be examined further in the literature reviews and studies, and work done by Richard Florida, economist and author of *The Rise of the Creative Class*.

This section reviewed the different services and amenities that contribute to QOL values, and the corresponding effects on employee migration. Prince George benefits from many QOL factors, including low housing costs, short commutes, and extensive outdoor recreation opportunities. While these factors are positive for potential residents, by helping to attract new employees to the area. However, there are a number of QOL factors classified as 'push factors,' and potential deterring new residents from moving to Prince George. It was noted that before QOL can be promoted by economic development officials to potential businesses, they must identify the elements that constitute QOL and their

relative importance of each to each individual businesses targeted. The second issue to note regarding the current strategy of using QOL to attract new residents and economic development, is that the responsible agencies or local municipal government does not influence some of the QOL factors being marketed and emphasized in recruitment efforts by employers.

The next section will discuss the specific types of QOL components being recognized as important amenities and factors in attracting the 'Creative Class' or knowledge workers with specialized skills. The 'Creative Class' comprises professionals, engineers, and skilled workers that Prince George economic development agencies and employers are being challenged to attract and relocate to Prince George. This practice is a significant issue to help address the skilled labor shortages and recruitment challenges in the region, as well as achieving economic development objectives to diversify the local economy. This section will also help to identify some potential QOL factors that can be influenced and improved by local government and economic development agencies, by investing into the community as an economic development strategy.

"Rise of the Creative Class" and Implications on Economic Development

The Rise of the Creative Class has been described by Glasier, 2004 as the most popular book on regional economies in the last decade. The book has received awards and acclaim from sources such as the Washington Monthly, Harvard Business Review, Money Magazine, the Phoenix, New Times, and Entrepreneur.com. While some critics of his work have come from economically conservative groups like the Manhattan Institute, anti-

immigration and homophobic groups have described Florida's thesis as an attack on bigbox business oriented development strategies, suburban lifestyles, and on 'family values'.

Florida's thesis began in 1999, with a series of studies to answer the question, 'How do you choose a place to live and work? From this study he developed the concepts and ideas to support his thesis concerning the rise of the Creative Class. Florida suggests that 'creative professionals' of the highly educated, well-paid segment of the workforce work in knowledge-intensive industries such as technicians in medicine, scientific research and healthcare, business management, financial services, and high-tech represent a high level of human capital. This group of people, contribute more than intelligence or computer skills, they add creative value. Florida suggests that creativity is increasingly valued by firms and organizations for the results that it can produce and by individuals as a route to selfexpression and job satisfaction. As creativity becomes more valued, the creative class grows and becomes more influential. Florida 2002, describes characteristic of the creative class in that its members engage in work whose function is to 'create meaningful new forms.' The super-creative new class includes scientists and engineers, university professors, poets and novelists, artists, entertainers, actors, designers, and architects as well as the 'thought leadership' of modern society: nonfiction writers, editors, cultural figures, researchers, analysts, and other opinion makers.

It is noted that this group includes industries that have been identified in Prince George as below the provincial average in the industry sectors employment. The Creative Class possesses considerable economic power as a result of its earning potential. Florida (2006)

Economy, roughly 30% of the entire US workforce, accounting for \$2 trillion, nearly half of all wages and salaries. In Canada, 3.8 million creative workers account for 25% of the workforce. It is not surprising that regions with large concentration of creative class members are also among the most affluent and growing economic regions.

Florida (2002: 3) also states that most civic leaders have failed to understand that what is true for corporations is also true for cities and regions, "Places that succeed in attracting and retaining creative class people prosper; those that fail don't." This view is echoed by Baris (2003: 42) who states: "Ensuring that creatives are welcomed, by extension, becomes the new task for cities. Thus, the old mode of people moving to follow jobs is turned on its head. In order to compete in the new race for talent, cities must restructure themselves for the Creative Class's needs much as companies have already done."

The question of why some habitats become destinations for the creative while others do not is similar to the economic term of 'low entry barriers'. As firms can easily enter new markets and regions, the same theories would apply for people and places where newcomers are accepted and fit in the social and economic networks. These are identified to by Florida, 2002 as 'plug and play' communities, where anyone can adapt and adjust quickly, where people can find opportunities, build support structures, be themselves, and are places likely to attract a greater number of talented and creative people. Talented people seek environments demonstrating acceptance and opportunity to differences and diversity. Florida (2002) suggests that cities and regions that attract lots of creative talent are also

those with greater diversity and higher levels of quality of place. This being location choices of the creative class based to a large degree on their lifestyle interest, and these go well beyond the standard 'quality of life' amenities that most experts think are important.

These lifestyle interests include nightlife and entertainment such as music venues, neighborhood art galleries, performance spaces, and theatres. The Creative Class seek out tolerant, diverse, and open communities, rich in amenities that allow them to maintain a work-life balance with demanding work schedules. Florida and Gates (2005: 131) argue that;

One of the primary indicators of these diagnostically critical conditions of openness and tolerance is the conspicuous presence of gays and lesbians, characterized here as the 'canaries of the creative economy', because of the way in which they signal a 'diverse, progressive environment'.

They also favor active, participatory recreation over passive institutionalized forms and value active outdoor recreation very highly. The Creative Class are attracted to places and communities where outdoor activities are prevalent, including traditional ones like hiking, bicycling, kayaking, and more adventurous ones such as trail hiking and snowboarding. In contrast to the economist Thomas Friedman's view, the world is flat, and people can 'plug in and play' anywhere, Florida suggests that the Creative Class migrate from major city to major city that have the desired amenities and QOL values. The Creative Class is also described by Florida as likely to reject suburbia, with its 'generica' of chain stores and malls; instead, they look for 'organic and indigenous street culture' and a range of other features such as walkable streets, with a profusion of coffee shops, art and live-music spaces. As noted by Peck (2005; 6) "Creatives want edgy cities, not edge cities."

As cities such as Seattle and Austin thrive and attract the Creative Class, many other cities are failing to adapt to the demands of the current knowledge economy and creative age. It is not through lack of effort, but rather that civic leaders are doing the limited things they think can spur innovation and economic growth. To this Florida (2002: 10) states;

But most of the time, they are either unwilling or unable to do the things required to create an environment of habitat attractive to the creative class. They pay lip service to the need to "attract talent", but continue to pour resources into recruiting call centers, underwriting big-box retailers, subsidizing downtown malls, and squandering precious taxpayer dollars on extravagant stadium complexes.

Florida refers to this as 'institutional sclerosis' as a symptom of places that grow up and prosper in one era, yet are challenged and often unable to adapt to the new organizational and cultural patterns, regardless of the benefits. The question then becomes: 'How would Prince George become a truly creative community - one that can evolve and prosper in this emerging age?'

Florida asserts that any urban center has an opportunity to 'turn it around' if they possess the essentials - a good university, 'authentic' neighborhoods, knowledge-intensive based industries, QOL, and cultural amenities that are important to the Creative Class. In the example of a city working to evolve is Providence, Rhode Island, Florida suggesting the strategy in creating 'a whole new model of university-community redevelopment'. Universities can play a leading role in the adaptive reuse of older urban facilities, "renovating the downtown core and surrounding areas as a seamless work-live-learn-play environment". (Catalytix and Richard Florida Creativity Group, 2003:2) Florida's critics include Malanga (2004: 45) who point out that "the production of authentic neighborhood cultures through deliberate public policy interventions is a daunting task, if not infeasible".

In this respect, the Florida creative-cities script identifies the nature of the challenge, the necessity for action and mobilization, and a warning to civic leaders. The warning by Florida is the threat is becoming a global one: competition for talented workers is no longer a domestic issue, and the next threat on the horizon is an international 'flight of the creative class' (Florida, 2005b). Florida informed the readers of *Money* magazine that the talent war is globalizing and cities in Australia, Canada, Ireland, New Zealand, and Sweden will increasingly be the ones to watch (Florida, 2005b).

This section reviewed and summarized the work of Richard Florida in identifying and defining the group referred to as the 'Creative Class.' As described, this dynamic group represents the group of professionals and specialized knowledge workers sought by Prince George employers to attract and recruit. This group presents enormous economic influence, due to their earning potential and impact on the success of employers and the regional economy. Discussed were the issues and challenges involved in attracting this group, and the types of amenities and services Prince George features, that are appealing to this group. Some QOL values are described that can be reviewed to help identify any 'shortfalls' in amenities or the strategy designed to attract this group of knowledge workers. The strategy that Florida suggests is a focused approach to the types of investments made into the community, and the group for which these investments are intended. It cannot be determined as a certainty that this strategy will provide complete success in addressing the region's recruitment challenges and to economic development, but it provides ideas and alternatives to be considered as a long term development strategy for Prince George.

Recruitment and Retention Issues in Prince George, BC

This section will review research on recruitment and retention issues in Prince George, BC. In 2005, Laura Way conducted a research thesis on recruitment and retention challenges as part of her Masters Degree in Political Science. I will review and outline some key points in her research and discuss the recommendations that have a resulted from her study.

Way tested the idea that Prince George experiences the same recruitment and retention issues as other small resource based communities. Despite being the largest city in northern BC, Prince George has faced ongoing challenges in recruiting and retaining professionals in the community. Way (2005:1) describes the situation, "While shortages of health care professionals received the most attention, a range of other professional positions remained difficult to fill, including university professors, lawyers and city planners". An example of the health care shortage occurred during December 2002, when the special care nursery at Prince George Regional Hospital was closed due to staffing shortages (Hoekstra, 2002). The central questions that were examined were: "how do new residents perceive Prince George; and what community attributes are they looking for?" (Way, 2005: 4). Way also tested which pull and push factors were consistent with the types of communities, and conducted a survey of new residents as well as human resource professionals. The results of the study confirmed that Prince George shared many of the pull and push factors of many smaller resource-based communities.

The pull factors identified by Way (2005) as having a positive effect for newcomers include employment as the primary reason for moving to the city. The natural setting of

Prince George is considered a positive geographical characteristic. Community size is considered a positive pull factor and negative push factor. The smaller city size, when compared to major urban sprawl in other regions, provides the ability to purchase housing and also allows for short commutes. The size of the community is considered a positive factor for young families who are raising children. The presence of post-secondary institutions is another favorable pull factor for recruitment for since it enhances the ability to continue further education and provides an option for their children. Listed below is the top 10 positive pull factors identified in Way's survey results.

	Relative Score	
Attribute	Rank	Score
Employment Opportunities	1	205
Proximity to Nature	2	56
Presence of Family and Friends	3	53
Size of the Community	4	52
Friendly nature of the community	5	47
Commuting Distance	6	45
Post Secondary Education	7	47
Spouse already in Prince George	8	38
Year Round Recreation	9	37
Good Place to raise children.	10	35

The undesirable characteristics of Prince George are those factors that tend to drive current and potential residents away. The primary push factor for Prince George determined by Way (2005) was the poor air quality, with 78 percent of survey respondents ranking it among the top five undesirable characteristics. Other push factors that received consistently high rankings were: climate, isolation from other major centers, and the economic attributes associated with a resource-based economy. The relative small size of the city also presented some negative perceptions that included the lack of shopping options as a source of dissatisfaction. Way states (2005:92), "It was important to note that respondents

who have families evaluate Prince George's attributes on the basis of how they think Prince George will meet their family needs". New residents who were single had other concerns. Way (2005) discovered that single professionals found it difficult to integrate themselves into the community; only 30 percent of single respondents were satisfied with their social interactions compared to 76 percent of married respondents in the study. It was determined that securing an offer of employment is often the first step to relocating to Prince George, and new residents placed varying degrees of weight on community factors.

The conclusions from Way (2005) identify a number of themes. Resource based communities have difficulty in attracting not only professionals that are experiencing country wide shortages but also other professionals, this challenge is often related to the inherent community characteristics. The second theme is that the reasons individuals move to, and from, resource based communities are push and pull factors related to overall QOL (Way, 2005). This is reflected in the comment that the ability to enhance pull factors, or resolve push factors, that are not directly related to the employment are largely outside of an employers control (Way, 2005). The third theme from the study suggests that the newcomer attitudes depended on demographic factors. It was recommended that when marketing to new group of professionals, employers will need to emphasize other characteristics of Prince George, rather than relying on its reputation as a good place to raise children as only 33 percent of the newcomer respondents had children (Way, 2005). While HR professionals in Prince George firms continue to emphasize community benefits in the recruitment process, their organizations continue to rely on the 'golden handcuffs' of a good salary and benefits package as a retention technique. This strategy was found to be

inadequate in addressing the recruitment and retention challenges in Prince George. Way, (2005:114) commented, "This thesis demonstrated that this approach is insufficient. Retention strategies must also incorporate broader QOL concerns."

This section outlined the specific qualities that are evaluated as both positive 'pull' factors and negative 'push' factors that attract or deter potential new residents to relocate to Prince George. This work was vital in providing an outline of the specific issues that can and need to be addressed to help improve and develop the local economy as a way of attracting new businesses and residents. Reinforced were the ideas that employers continue to rely on traditional means of attracting employees with compensation and benefits while it was found that many factors used to consider relocation to Prince George were not employment related. Also noted were the concerns that this strategy is inadequate to address the recruitment challenges and that QOL concerns needed to be addressed by the local government and community. It is important to consider this information and address the issues in the recruitment and economic development strategies used by employers, local governments and economic development agencies.

Section IV Literature Review

Determinant of Location
Quality of Life and Location Decisions
Role of Quality of Life in Technology Based Firms
Impact of Quality of Life on Recruitment and Retention
Quality of Life in Economic Development Strategy

The literature review section is a summary of studies on the impact of QOL on the location decisions of firms, and the impact on recruitment and retention of key personnel. The

literature review includes traditional industrial location theories, and the role of QOL in technology based firms, on recruitment and retention, and in economic development strategies. This review was completed to build an understanding of the theories and studies on QOL issues and the impact on recruitment and retention.

Traditional industrial location theories have been based primarily on cost minimization that includes resource such as land, labour, materials, and transportation costs. Locations are selected by firms based on profit maximization and economic theories. Most of the early studies on industrial location focused primarily on the minimization of transportation costs (Blair and Premus, 1987) which is still prevalent in the process for many industrial firms today. Additional cost minimization factors included taxes, property values, construction costs, local regulations, and strength of local labor unions. Indirect costs and benefits may have significant impact on an organization's bottom line and operational success, some of which are government services, availability of infrastructures, climate, and local environmental quality. Past studies have been based on tangible and direct cost factors versus the intangible and indirect factors that are associated with QOL and public services that with technological, environmental, and economic change has become more significant.

The decision to expand or relocate are understood as strategic decisions by an organization and are part of the larger corporation planning process (Blair and Premus, 1987; Cohen, 2000). Early studies on industrial location decisions highlight the cost-minimization along with profit maximization considerations as the primary factors in the decision process. An example of which is; at the locations where the combined costs of transportation of raw materials to the location and transportation of the output to the market was at the lowest

point. This is in addition to the location where the discounted net present value of its production is greatest. This profit-maximization assumption is followed in many locational decision processes today. The next step in the location decision is to compare the potential locations against 'must have' and 'would like' item lists of amenities. The 'must have' items are essential to the firm operations and strategic objectives, while the 'would like' list includes desirable factors but not instrumental to firm operations. This process also includes consideration of negative factors as part of the process. It has been found that when the major factors or 'must have' items have been met in locations, other amenities and benefits are considered in the location decision. Schmenner (1982) noted that when several locations match on the most important attributes, seemingly idiosyncratic or insignificant factor can play a large role in deciding the ultimate choice. The final stage of analysis considers the attributes of particular communities and/or alternative sites.

Quality of life factors are generally of lesser importance for manufacturing sectors, which are most sensitive to the traditional economic factors of locations. However, economists have become aware that many decisions, including location, are not based purely on profit maximization (Walter, 1975). Today, an increasing number of industries can operate successfully in a variety of locations, as a result of changes in production and communication technologies, possessing cheaper and faster forms of transportation, and due to changes in the industrial composition of economic regions. It is noted by Schmenner (1979) that firms often consider several low-cost sites, and a seemingly inconsequential item tips the scale in favour of one site. It is imperative for local planners and economic development groups to understand and highlight these in their concern to create

employment and economic opportunities in the region. Business relocations are frequently cited as the most direct way to enhance a region's economic base. Furthermore, when relocations do occur, the vast majority are over very short distances, often within the same community (Schmenner, 1982).

Determinant of Location

The most comprehensive analysis of industrial location determinants have been based on surveys because they allow researchers to probe the significance of long list of variables, some of which may be qualitative (Blair, 1987). For instance, QOL is important to many firms, but it is a concept that is difficult to quantify. Morgan (1964) reviewed locational studies prior to 1963, and of the earlier industrial location literatures lead to the conclusion that traditional location factors were dominant influences on industrial location, negating the influence of non-economic factors. QOL and other nontraditional factors were not significant in these earlier studies (Blair, 1987). Blair considered the most comprehensive survey of high technology location decisions was based on the work of Premus for the Congressional Joint Economic Committee of the United States (Premus, 1982). The study is based on the survey response of 691 high tech company executives, where the dominant considerations were identified as availability of technical labour and the cost of the labour pool. Schmenner (1980) also interviewed executives from 29 high tech firms and found that they were oriented to technical expertise in the labour market. Other location factors considered important included community attributes (QOL factors) such as good schools, space for expansion, local transportation, recreational, and household amenities (Blair, 1987). High Technology firms have been a growing sector in the economy and are highly

prized by economic development officials (Salvensen and Renski, 2002). High Technology firms are usually defined by large expenditures in research and development with a percentage of employees who are technicians, engineers, and scientists. The availability and cost of technical labour is perhaps the most important determinant of location for high-tech firms, followed by proximity to universities (Schmenner, 1982; Premus, 1982). Research and development firms, a subset of the high tech industry, are highly sensitive to QOL issues, primarily caused by their need to recruit and maintain knowledge workers (Ritter, 1990). Premus (1982) also identified taxes, cost of living, and transportation as important regional determinants of high-tech locations.

In general, the executives of high technology firms listed QOL factors ahead of traditional business location factors. A review of past studies reveal that industrial location decisions were primarily based on access to market, labour market, transportation costs, and access to raw materials. To a large extent this is still the most influential for industrial and manufacturing firms. The list of location decision factors expands and shifts for technology based firms, to include business climate, labour skills, and local physical infrastructures. These nontraditional location factors dominate the location choices of high tech industry firms; thus indicating that as the province's industries continue to shift to more advanced technologies, the degree of reliance on nontraditional locational factors including quality of life will increase.

Role of QOL in Technology Based Firms

Four approaches to job creation has been sought by public and private economic development organizations; attraction of new businesses, expanding existing businesses, retention of existing businesses, and formation of new businesses. Blair and Premus (1987) concluded from their extensive review of literature on the major factors influencing industrial location, advised the United States government to focus its efforts on improving the overall locational attractiveness of regions, emphasize long-run tax policy, management-labour relations, and quality of life factors. Industrial development would largely be a by-product of an improved, overall business climate, and a better community in which to live.

This new approach recognizes the increasing importance of QOL in business location decisions (Love and Crompton, 1999). QOL has been reported in studies as being influential for high technology industries companies involved in research and development, as well as firms that employ highly skilled workers in information and knowledge-based services (Schmenner, 1982). This emphasis on QOL recognizes the economic success of business is dependent on a quality workforce. QOL is not only important in relocation, expansion, or initiating decision, it is also important in employee retention, as it is expensive to go through the recruitment process particularly for key personnel (Love and Crompton, 1999). This view is supported by Taylor (1987) who asserts the quality of employees' lives have a direct impact on an employer's bottom line through absenteeism, loyalty, turnover, productivity, and health care costs.

In a study by Love and Crompton 1999, the results of research illustrated that the highest ranked factors among sampled firms were labour costs combined with daily living concerns. A relatively low ranking of QOL was not anticipated in their study, given the increasing focus in business relocation literature on the role of QOL in business location decisions. Their study also provided insight into the characteristics of companies that placed a relatively high importance on QOL;

- Small companies employing fewer than eight employees placed significantly higher importance on QOL compared to large companies with 88 employees or more.
- Footloose companies, defined as businesses whose financial performance is relatively
 independent of location decisions placed significantly more importance on QOL factors
 as there is limited dependence on raw materials, natural resources, energy supplies, or
 local geographical market.
- Companies in which the primary decision maker relocated with the company place significantly more importance on QOL than those companies in which the primary decision makers were in a different location, ie head office.

The study of technology based firms provide a basis of information that are considered in the location decision of these companies. This is important to consider in the economic development of Prince George and the strategies that can be employed to attract such industries in effort to diversify the local economy. Also important is the influence of QOL in the location decisions of such firms.

Impact of Quality of Life on Recruitment and Retention

Glaser and Bardo (1991) conducted exploratory research for improving public sector economic development, through QOL investments to assist the business community's ability to attract and retain key personnel. Their study concluded that the business identity has substantial bearing on the values assigned to QOL factors for key personnel recruitment and retention. Across all business types, access to quality key personnel is an important component for business success and economic growth, yet the literatures concerning this relationship are limited. Public sector investments in QOL not only highlight the community as a more attractive place to live, but also contribute indirectly to economic prosperity (Glaser and Bardo, 1991).

A significant portion of the literatures on QOL as it relates to economic development indicate that issues of key personnel recruitment and retention are not specifically addressed (Graves, 1980; Porell, 1981). Although there are relationships between the attraction and retention of key personnel and an area's QOL, there is limited information about the relative significance of various QOL attributes. Furthermore, many employers have found it difficult to recruit, transfer, and relocate highly skilled workers to remote locations (Glasmeier, 1986; Harding, 1989). A warning was presented by Myers (1987) that cities must consider quality of life assessments seriously if they expect to remain economically competitive. Failure to improve local QOL can result in a situation in where businesses will find it necessary to pay special fees or 'disamenity incentives'. Access to better job opportunities is the primary draw of professionals to larger urban areas (Herzog and Schlotman, 1986, 1989). While the small town environment, fosters 'traditional values'

that are attractive to professionals wishing to balance work, family, and community (Daniel, 1993). Despite preferences for rural QOL benefits, most high-tech and knowledge professionals choose to live in large metropolitan areas (Glasmeier, 1986; Markusen *et al.*, 1986).

QOL in Economic Development Strategy

This section reviews the literature on arguments for using amenities as an economic development tool. Though limited, articles that review theoretical, survey, and econometric literatures on amenity-oriented firms location and employment growth indicate that there is little evidence that firms base location decisions solely on QOL factors. Gottlieb (1994) examines the evidence for the use of an 'amenities strategy' in local and regional economic development. Noting that such a strategy would differ from traditional economic development strategies in its focus on QOL as a key determinant of firm location. Most economic development officials believe that local amenities affect their ability to attract jobs, but few would make the provision of such amenities the centerpiece of their economic development program.

Business retention and expansion are pointed to in the literature as being a more effective economic development policy in comparison to business recruitment (Glasser and Bardo 1991). Economic development means business retention and expansion rather than employee recruitment to the vast majority of communities (Rancer, 1979; Rooney, 1986).

Glaser and Bardo (1991) suggest that the most obvious and important conclusion from their findings is the targeting of business type in a planned economic development strategy.

Recommendations are made to economic development groups to target efforts according to the business types, as the specific needs and values of businesses will affect the QOL for key personnel. It was also noted that established industries provide easier targets for further focus in the short term, while industry diversification offers long term protection against fluctuations in the local economy, but is consequently more difficult to target. This concept is a fundamental decision to the important economic planning considerations in business targeting (Glaser and Bardo, 1991). When local development agents are assessing key personnel recruitment and retention strategies, care should be taken to separate issues according to the extent to which they are addressable through policy action. Certain quality of life attributes are of a fixed nature. The best example in this study relates to local climate. Climate was a moderate priority for key personnel in selected businesses but was clearly unimportant to others. Since climate is fixed, localities with less than attractive climates may want to target businesses types that consider climate of lower priority.

These findings lead to some general conclusions for local government and economic development strategies. QOL is an important factor for retaining or recruiting key personnel and is arguably an important issue in economic development strategies. An initial assessment of labour force requirements to business needs is critical for an effective economic development policy. It is also interesting that the findings in the study indicated technological support from a university was a common theme for CEOs who were experiencing difficulties recruiting key personnel. This trend would further suggest that local regions would be well served by paying closer attention to QOL attributes that are considered significant, and that can be modified through a conscientious investment

strategy, such as local recreation and entertainment facilities in addition to public education and health services. Areas that are able to attract and retain skilled workers are most likely to be successful in the near future, and pools of highly educated and technically skilled labor can only be maintained in areas with an appealing QOL and favorable amenities (Schmenner, 1982). This point is reinforced by the more recent work of Richard Florida.

The literatures and studies conducted have for the most part, been limited to key personnel issues and personal QOL preferences of CEOs and their influence on locational decisions. Direct interviews with key personnel would likely provide further enlightening insight regarding their motivation for site selection and relocation. Further research is also needed in analysis of differences based on business types.

Section V Methodology Key Informant Interviews General Aspects of Retention in Prince George General Aspects of Recruitment in Prince George

The methods of research in previous studies have been a combination of extensive surveys to regional interviews. It was intended to use a sample of 10 firms/sources for interview to gain insight to the relative importance of public service (health, education, and recreation) and QOL in the recruitment and retention challenges of firms in the region.

There was effort made to include organizations of small, medium, and large sizes in the various industries to gain breath and depth in the sample. The purpose of this process was to explore the scope of QOL considerations by human resource managers in organizations.

A sample of 10 respondents was used to allow enough time for probing with some depth in the interviews, with a limited sample intended to be illustrative, not representative of the region's organizations and employers. By using a cross section of different industry types, this study was intended to gain a broad sense of HR managers' QOL awareness and considerations across industries in the region in their recruitment and retention efforts.

The sample of firms included in the interview process was selected from a list of organizations in the region. Using a list of 25 major employers in the city provided by the City of Prince George and a list of organizations in the community provided by Living Working Prince George. From this random list of employers, I began to contact the firms until I had 10 interviewees who agreed to be interviewed. This constituted the final sample in the study. I did not attempt to conduct a comprehensive survey of businesses from a large sample, due to both time and resource constraints by the nature of this research project. Respondents in most cases were either the head of the firm or the manager of human resources, were asked to identify the importance of QOL in the organization's recruitment and retention efforts. The interview process was used to obtain similar information from multiple individuals. Using a structured interview guide, containing both structured and unstructured questions, both open and closed questions to enhance the reliability of the results (Hedrick, 1993). Each interview contained general questions around the following themes: 1) does their organization have concerns regarding employee retention and what methods are used to retain employees; 2) does the organization have concerns regarding recruitment and what methods and costs are associated with employee recruitment; and 3) do OOL issues in Prince George affect your employee recruitment and

retention efforts. There were two benefits to the interview method used: first, the type of data received was not limited by the constraints of survey responses inherent in its inflexibility and second, the questions allowed the respondents the opportunity to add additional issues and information to the interview process (Babbie, 2001). Again, the purpose of the study was to gain insight into the relative importance of QOL in the recruitment and retention of organizations in Prince George.

Interviews were conducted to obtain employer perspectives on the relative impact of QOL on the recruitment and retention aspects of their human resource management practices. Interviews were conducted primarily with human resource managers of the companies and organizations. The research conducted through interviews included ten participants in Prince George. The organization participants include the following evaluative variables; Public and Private Sector:

• Five (50%) of the participants were from public sector organizations and five (50%) were from private sector organizations.

Employers grouped by number of employees:

- Two (20%) of the organizations were small employers with less than 30 employees.
- Four (40%) of the organizations were medium employers with less than 150 and more than 30 employees.
- Four (40%) of the organization were large employers with more than 150 employees.

Organizational Structure:

• Two (20%) of the organizations have a private ownership structure.

- Two (20%) of the organizations are non-profit organizations.
- Three (30%) of the organizations are public/municipal organizations.
- Three (30%) of the organizations have public ownership structures.

The following section will review the responses and common themes regarding employee retention in Prince George. The interview questions attempted to identify any existing concerns in employee retention and methods of employee retention.

General Aspects of Retention:

A question that was included in the interview related to issues of retention.

5) Does your company have concerns regarding employee retention?

- Number of respondents who said yes: 4
- Number of respondents who said no: 6

Some of the common concerns regarding employee retention drew upon the trend of an aging workforce; this creating the need to find and develop replacements for the aging 'Baby Boomers'. Other issues in employee retention include the casual or part-time employees who pursue full-time employment outside of the organization or leave because of relocations of their spouse or partner. This has been spurred by the improvements in the economic health of Prince George and recent declines in unemployment where there are more employers looking to fill permanent full-time positions. Other concerns expressed included the issues of supply and demand in the skilled professions where there are opportunities in many regions outnumbering the qualified candidates. This has resulted in

positions that are unfilled for extended periods of time; a particular issue in the health cares professions.

A specific aspect of employee retention depends on the industry type. For example, some industries or companies employ educated professionals who use the experience in Prince George as a development stage and pursue higher levels of training and employment outside of the region. Respondents in both the non-profit and public sector employers expressed this view. The respondents with private ownership structures indicated that employee retention is not a concern, and turn-over is moderate or below industry average. This view was shared with respondents of medium sized employers; in contrast to the respondents of the large employer segments who were more likely to have retention concerns due to employee retirement issues. One respondent stated, "Its getting them up here that is the hard part, once they come and see what we have to offer, retention is not as challenging."

6) What are methods used to retain your employees in the company and in Prince George?

Table 1.2 Listed below are some methods used by respondents to retain employees

Method	Frequency Mentioned	
Competitive compensation and benefits package	100%	
Market comparison of compensation for industry positions	60%	
Bonus Program & matching RSP contributions	60%	
Employee performance and service recognition	30%	
Health and wellness programs	20%	
Efforts to engage participation and gain employee feedback	20%	
Encouraging work/life balance	20%	

Providing competitive compensation and benefits package is the primary retention tool for employers. A respondent had indicated that this was further developed to enhance the retention process by creating accommodating working conditions, and providing supplement income to retain specialized professionals in the region. The associated costs to some employers have increased as a result of higher compensations and benefits to retain skilled employees. In general, it has been the 'golden handcuffs' that are used by employers to retain employees. Way (2005) suggested that this strategy was found to be inadequate to address the long term recruitment and retention challenges in Prince George. It is also being identified as an increasing source of pressure for employers who are required to provide higher levels of compensation and benefits to retain employees, increasing pressure on the financial resources and management of the organization.

7) Are there common reasons employees leave your company and Prince George?

Table 1.3 Listed below are some reasons employees leave their employers

Reasons	Frequency Mentioned	
Employees retiring and leaving	40%	
Employees are recruited by other industry employers	30%	
Employees leaving Prince George for more 'amendable' locations	20%	
Limited opportunities of industry in Prince George	20%	
Nature of the industry, high work related turn over	20%	
Employees leaving with spouse employment relocations	20%	
Economic & Employment prospects of industry in Prince George	20%	
Employees leaving for higher levels of professional development	20%	

The responses to this question highlighted the growing concerns with respect to employee retirement and its effects on company staffing. This is particularly difficult for some sectors who are also faced with high competition from other regions that are considered more

'amenable' based on weather and climate, location, and city amenities such as Kelowna, Kamloops, Vernon, Victoria, and Vancouver. Another common concern is the increased level of competition for employees among employers within the same industry in Prince George, this has been as a result of decreased unemployment rates. The employees who left their employers as a result of spouse relocation were commonly secondary income earners in their families and/or part-time working mothers.

The respondents with concerns regarding retirement were primarily from large and medium organizations in both the public and private sectors. Respondents who experienced employees leaving to join other firms were mostly from the private sector, while employees leaving for more amendable locations occurred in both the private and public sectors. Some of the reasons identified for employees leaving were industry specific that included high work related turnover rates, limited advancement opportunities in Prince George, and leaving to pursue further career development.

The respondents identified some prevalent methods that are used to retain employees. These methods have been focused on the compensation and benefits package for employees. Despite these efforts, employers continue to encounter employee retention challenges. One common theme in employee retention is workforce aging and retiring 'baby boomers'. This issue was a concern for employers in both private and public sectors, and more relevant to large and medium sized organizations. The decreased unemployment rate in Prince George was also suggested as an important employee retention factors

challenge. The responses suggested that the primary influence on employee departure related to improved employment opportunities in other industries or regions.

The following section will review the responses and common themes regarding employee recruitment. The interview questions attempted to identify any existing concerns in employee recruitment, methods of employee recruitment, and costs associated with recruitment efforts by Prince George employers.

General Aspects of Recruitment:

8) Does your company have concerns in regards to employee recruitment?

- Number of respondents who said yes: 9
- Number of respondents who said no:

The interview responses to this question clearly indicate that employee recruitment is a challenge for Prince George employers. Respondents with recruitment challenges were from all respondent variable groups: large, medium and small organizations, public and private sectors, as well as all organizational structures (private ownership, non-profit, public/municipal and public ownership organizations). Similar to employee retention issues, recruitment efforts have been heavily impacted by the changes to the local and provincial economy as well as the local and provincial employment rates. Decreasing rates of unemployment translate to higher levels of competition among employers for qualified and trained candidates. A common theme among respondents is the trend to extend employee recruitment for specific industries and professions outside the community, a direct result of the limited pool of qualified candidates in the region.

The challenge to fill vacancies within the organization have required the firms and HR managers to consider alternative options. These options include creating developmental roles or 'fast track programs' that allow employees to train and gain experience in entry/mid level positions before they can fill the vacated postings. Some medium and large organizations have the resources to encourage movement employees within the organization used this method. A second option respondents identified for dealing with recruitment challenges involved recruitment efforts in other regions of BC and western Canada such as Alberta, Saskatchewan, Manitoba, and the Yukon. This method was most common in medium and large organizations and respondents in both public and private sectors. A respondent in the private sector considering even further options in recruitment, "I have been thinking about recruiting outside of Canada." This option is most available to organizations that employ skilled, educated professionals such as those in the medical field, as commonly evidenced by the migration of South African doctors to northern and rural locations of British Columbia. This option is also available for firms in the engineering industry, since the training and designations earned by professionals are frequently recognized internationally. To further comment on this issue, the same respondent also suggested that "everyone with skills and experience is working" at this time unless experiencing extraordinary circumstances. This comment clearly reflects the reduced unemployment conditions in the region and across the province.

For respondents, employee recruitment was clearly a greater concern for employers in contrast to employee retention concerns, as suggested by the interview response. Some of the challenges faced by centers on overcoming the more common negative stereo types of northern BC. Some of the negative stereotypes include; a city full of loggers and 'red necks', isolation and remoteness, and concerns of weather and climate. These concerns emerged frequently among the interviewees across industry types. One respondent suggests that potential candidates being recruited from the lower mainland have "no idea where Prince George is."

The following section will review the responses and common themes regarding employee recruitment. The interview questions attempted to identify employee recruitment methods, and if this method varied depending on the type of position that is being recruited.

9) A. What are methods used to recruit your employees to the company and Prince George?

Table 1.4 Listed are common methods of recruitment used by employers

Methods	Frequency Mentioned	
Local media - The Citizen paper	90%	_
Company/Organization website	70%	
National Advertising (Print Media- National Post)	60%	
Industry/Trade Journals	40%	
Recruitment Fairs (local & external)	40%	
Employee Referral program	30%	
Recruiting from competitors	30%	
Foreign Recruitment	20%	
Head Hunters & National Placement Agencies	20%	

The above list identifies the methods used by respondents to attract candidates for vacancies in their organizations. The most prevalent methods were local media and the company website. These two options were used for all job types and entry/administrative

positions within the organizations. The advertising undertaken by respondents would vary in regions, but commonly is concentrated on BC and western Canada.

There were some common comments made by respondents on non-employer related factors that impact recruitment efforts as well as some of the challenges in the process. Half of the respondents observed that they are aware of competition from other regions for skilled, or trained professionals. Factors such as climate and perceptions of remoteness remained a potential challenge for recruiting candidates from outside the central interior of BC. Three respondents also indicated that these factors were dependent on the candidate's current location. If the candidate was from the BC interior, it was not a concern; however, if the candidate was from the lower mainland or outside BC, it could create a potential challenge. A manager in the public sector indicated a large portion of the relocation decision is, "not associated with the work or compensation and the focus is on the location, amenities, and climate." This comment was supported by another respondent who suggested that when recruiting outside the BC interior region, a candidate's decision to relocate to Prince George, "Is not 100% about the job or organization and it is more about the area and location and what it has to offer the individual and family". These comments were made by respondents in large and medium employer types. A respondent in the small employer category stated, "compensation is not enough" to attract the necessary skilled professionals as a stand alone incentive. This statement is made in the context of competing with other employers and regions, when dealing with the challenges of limited resources to provide cash incentives.

Some indirect costs and methods associated with the recruitment process described by respondents include higher training costs for new employees with less experience and training than is required. This calculation also includes the use of developmental roles and positions, to provide the required training and experience for employees to fill specific positions within the organization. This situation results from an inability to find candidates with the required training or experience. One respondent suggested that the desired employee segment being targeted are the older workers, or 'baby boomers', while the applicants received are from younger, less experienced candidates. When employers are marketing their employment positions in different media sources, whether locally, regionally across the BC interior or nationally, there is a pronounced emphasis on city amenities such as the college and university, multiplex, hospital, as well as outdoor recreation, cost of living, and the regional lifestyle of affordable housing, limited commute, and small town lifestyle with city amenities. This was a common practice among nine of the 10 respondents. These factors while important are not influenced by the employers and are recognized as the positive QOL that employers rely heavily on to attract employees to the city.

B. Does this process vary for different levels of staff and management?

The methods to recruit management, administrative staff, and specialists are different for 7 of the respondents, with 3 respondents indicating they use the same methods for all positions. The three respondents who used the same methods of recruitment for all positions were from the small and medium size organizations.

The respondents who indicated a different strategy of recruitment for administration, directors, and managers all used marketing methods outside the region to ensure extensive coverage of the province and western Canada. This tactic was also followed on a national level for executive positions. The level of marketing also involves higher costs to advertise via national newspapers such as the *National Post*, as well as using 'headhunters' or hiring agencies to fill executive level or specialist positions. Primarily large organizations use this method of recruiting executives in both the public and private sector. The challenges in recruiting skilled and trained professionals are compounded when recruiting from outside the BC Interior region.

A large contributing challenge in recruiting from outside of the region is the unfamiliarity of candidate in the city's locations, amenities, and QOL. Compounding to this challenge is the fact that executives, specialists, and trained professionals have the capacity and ability to relocate to regions of their preference and are not limited by employment opportunities or regions looking for skilled professionals in the current employment market.

10) What are the costs to recruit and fill a position within your organization? Is this cost different for different levels of staff and management?

Many of the respondents have indicated that there are no accurate measures of the cost to the organization in recruiting local and external candidates to the region. The most common costs were associated with the marketing and advertising in local and external print media, such as in trade journals, the *Prince George Citizen*, *Vancouver Sun*, *Province*, and *National Post*. Estimates for this recruiting method range from \$300 to \$3000. Other

recruitment expenses involved the costs of travel and relocation, training, and development. One respondent in the public service sector estimated the cost to recruit is up to half the candidate's salary for the position. Another respondent in the private sector suggested the cost to use a 'headhunter' to fill a management position is as much as \$20,000 in fees. Additionally, some respondents in the private sector are relying more on referral and 'word of mouth' recruitment, where bonuses are paid to employees who refer a new hire to the organization. The referral bonus pay ranges from \$500 up, depending on the position and experience level.

The recruitment and retention efforts by employers have been the traditional methods of using incentives and compensation. While it has been effective to some degree, the challenges in recruitment by employers in the region still persist. Despite not having accurate estimates of the real costs in recruitment efforts, the estimated direct and indirect costs suggest a large amount of resources in time and money to fill essential positions. The most difficult positions to fill are the management, specialists, and executive positions. These positions are the most costly to recruit and are often remain unfilled for extended periods of time. This indicates that employers in the region, regardless of employer size, sector, or organization type, are in a difficult position to attract the candidates who are higher skilled and/or have specialized training. The controllable factor that employers possess in this effort is the compensation and benefits packages that are offered to the candidates. All other factors are uncontrollable, and are influenced by local and municipal government in terms of public services and infrastructures, recreational facilities, and services of the region. The region's housing costs, cultural, arts, and entertainment

offerings including restaurants and night life are also influenced by private investment. It was also identified that employers place a much higher rate of concern in recruitment as compared to retention. This was identified in the 30 percentage higher number of respondents who indicated concerns regarding recruitment compared to concerns regarding retention.

Question 16) Do QOL issues in Prince George affect your employee recruitment?

Number of respondents who said yes:

Number of respondents who said no:

The majority of respondents indicated that QOL issues in Prince George has a material effect on their employee recruitment efforts. Some of the QOL factors have been discussed in the previous section, and how these factors are heavily marketed with the recruitment efforts in regional and national media. The literature review indicated that higher skilled employees and the technology industries are more sensitive to QOL factors, in the location decisions of businesses and migration of employees. This idea was supported by Richard Florida's research on migration patterns of the 'Creative Class', and the focus on QOL factors. A respondent in the public sector supports this idea and view in "recruiting a group that can live and work anywhere." When the recruits make the decision to relocate and have the ability to move to any area, the focus on QOL factors are magnified in importance.

The limitation to this question is that it does not measure the degree of importance QOL factors are on recruitment efforts. The QOL factors that impact the recruitment efforts

include both the positive and negative factors. The negative factors are considered in some aspects more important then the positive factors. Another respondent in the public sector stated that QOL is "clearly" important, and as simple as the "air quality" in Prince George. The Prince George air quality was identified by Way (2005) as the most important 'push' factor in her study of recruitment challenges in Prince George. Respondents also commented that QOL was becoming more of factor and that they are more aware of this in their role. One respondent indicated that the overall QOL in Prince George was overall positive, citing outdoor recreation, housing affordability, improved art and cultural facilities, but also identified air quality as the primary negative issue. Skilled professionals exhibit a higher sensitivity to QOL factors and public services, as indicated by the respondents and supported by the information in the literature review.

17) Do QOL issues in Prince George affect your employee retention?

- Number of respondents who said yes:
- Number of respondents who said no:

The responses to this question indicates that employee retention and QOL issues are less of a factor in retention when compared to recruitment. These responses are in line with the percentage of responses with concerns regarding retention challenges by employers. It was discussed that employees generally do not leave because of lack of services, and the reasons for leaving are primarily for career and employment reasons. A few comments were made by respondents from the public service sector about the issues of new families making connections and developing 'roots' in the community. It was noted that new employees who relocate to the region and do not have family or create a social network will leave much quicker then new employees with family in the region or who created local

social connections. Another respondent added that one of the most critical factors is the challenge of integrating the new employee's spouse into the community. Important considerations in this process are the region's cultural and social aspects, since spouses generally are very conscious of these two factors. A further hurdle is the influence that children have on retaining employees, and if they are happy with schools and activities. This feature relates to the frequent comment made by respondents that Prince George is a 'good place to raise children'. Way (2005) had suggested in her research that when marketing to new groups of professionals, employers will need to emphasize other characteristics of Prince George, rather than relying on its reputation as a good place to raise children since only 33 percent of the newcomer respondents had children.

Some of the respondents who indicated that QOL were issues in employee retention cited concerns regarding the geographic location, including distance, and remoteness from major urban centers as retention factors. The distance from major centers is considered an issue for both professional and personal reasons, which include connection to industry and lifestyle benefits of living in a major urban center versus a small community. Major urban centers such as Vancouver and Calgary are the locations of corporate head offices, while the city amenities provide QOL and lifestyle factors that are not found in smaller centers. These lifestyle factors include more amenities in retail, restaurants, shopping, entertainment, nightlife, arts, cultural events, organized recreational and competitive sports, including professional sport franchises.

Section VI Interview Results and Findings

The interview responses by the managers and human resource managers in this section highlight some of the issues concerning the role QOL plays in employee recruitment and retention. It was found that employee recruitment is a challenge for employers in all sectors, of all organization size or structure. Contributing to this challenge are the strong economic conditions in the province and Prince George along with declining unemployment rates in the region. Employee recruitment was indicated by respondents as a much greater challenge compared to employee retention. The recruitment efforts by local employers greatly emphasizes the regional 'lifestyle' factors of outdoor recreation, low cost of housing and commute, city services, and amenities such as the university, in an effort to attract candidates to the region.

The challenges discussed in the recruitment process include overcoming perception problems of isolation, climate, and remoteness. Other challenges in recruitment discussed were competition between different employers and industries both within the region and among other communities and regions. The costs associated with the recruitment process are significant for all employer types and sizes, regardless of public or private sectors. The results and responses from the respondents suggest that larger organizations have a higher level of awareness of the challenges in employee recruitment and retention compared to small and medium sized firms. Large organizations have the resources and capacity in human resource management departments to address these challenges more appropriately. Small organizations are limited in its resources and capacity to address these challenges and are the most significantly impacted by the inability to use more resources to fill

specific roles in the organization and in having these positions unfilled for extended periods. Medium sized employers are aware of the challenges, while not as significantly impacted as small organizations, do not have the resources and capacity to address the issues in the same manner as large organizations.

The expenses associated in the recruitment process includes direct and indirect costs such as marketing, hiring 'headhunters', travel and relocation costs, training fees and financial losses in having the position vacant for extended periods of time. Yet the only controllable factor that employers have in this process is the marketing of the position, and the compensation and benefits packages that are offered to fill it. Wage rates for professionals are commonly set at national levels and tend to not vary significantly among local labor markets (Topel, 1986). Because compensation levels for professionals generally vary slightly between regions, professionals tend to distinguish locations according to quality of life factors rather then wage rates (Morgan and Sayer, 1998) The compensation is the primary method used by local employers to attract and retain employees in the organization. This identifying the shortfall in this strategy of using the 'golden handcuffs' and point to the level of success achieved in attracting the necessary employees to the community. While a significant factor in the recruitment process includes the candidates assessment of the OOL factors that will benefit the candidate in terms of city services, amenities, recreation, arts and culture, entertainment, and life style factors. Many of these QOL factors are subject to local government policy. It is important that local economic development policy be based on a foundation of understanding the QOL priorities and its effective use of local economic development resources.

The impact of QOL on economic development was not explored in the interview process, as most interviewes were human resource managers with the primary theme of the interviews related to recruitment and retention challenges. While there is evidence to suggest that public sector investments in QOL not only make the community a more attractive place to live but also contribute indirectly to economic development; further research focused on the economic development aspects of QOL would be beneficial for further contribution to the study.

The following section is a discussion of QOL, in recruitment and retention as well as the impact on the region's economic development strategy. Included is a discussion of how the literatures reviewed supports and differs from the research responses in the interviews. Recommendations are made to address these issues in Prince George with a description of the limitations of the current study and suggestions for further research.

Section VII Conclusion
Discussion of analysis related to the literature review
Results & recommendations
Limitations & further research

The purpose of this study was to determine the role and impact quality of life has on economic development and employee recruitment and retention. There exists a prevailing perception that a high QOL is available in rural areas and in Prince George, with some valued amenities such as outdoor recreation opportunities, less traffic commute, and lower cost of housing, all of which are constantly promoted in an effort to attract new employees to the city. There are other QOL factors that are more common in larger centers and

metropolitan areas, generally associated with threshold city sizes such as Vancouver and Calgary. Several perspectives emerge in the literatures that suggest how amenities and other QOL factors influence business location decisions through labour force preferences. Understanding the demand for their skills and services, knowledge workers including the 'Creative Class' are highly mobile and choose to live in amenity rich areas that display the locational preferences.

The interview responses strongly indicate that QOL factors affect their recruitment efforts regardless of employer type or size to varying degrees, with greater influence on potential employees from outside of the region. Also identified was the greater challenge in attracting the higher skilled employees in positions of management, executives, and specialists among the different organizations in the city. This is identified in the literatures that suggest that failure to improve local QOL can result in a situation in which businesses will find it necessary to pay special recruitment fees or 'disamentity incentives' (Myers, 1987). Cost of living, housing, culture, restaurants, and local area entertainment all have been shown to be important OOL issues that in some way are related to the recruitment of key personnel in literatures (Browning, 1980, Hack 1984, Levy, 1981). Entertainment measures are a composite including restaurants and cultural opportunities, all of which have overlap and significant appeal to the 'Creative Class' as suggested by Richard Florida (2005). The most significant finding regarding QOL in recruitment is in the policy perspective since QOL related investment to the local environment could be achieved in the short-run, although they will require public-private sector cooperation by local government and private organizations. In summary, business type and size has some bearing on the

values assigned to various QOL elements, while the importance in personnel recruitment and retention is consistent with few exceptions.

There are several perspectives pertaining to how amenities and QOL factors influence business location decisions through labour force preferences. The perspectives in the literature argue that local amenities can affect a firm's bottom line directly by lowering wages and reducing costly labor force turnover for key personnel related to local QOL (Granger and Blomquist, 1999, Gottlieb, 1994, 1995; Taylor, 1987). This view has been practiced for many decades prior to the shifts to technology or knowledge based economies. As Stein (1952, p. 3) argued in the design and development of the resource based town Kitimat, BC:

The purpose of Kitimat is the industrial success of the plant. That success will depend on the degree that workers are content, that they like living in Kitimat. Unless the town can attract and hold industrial workers, there will be continuous turnover and difficulty.... The workers must find Kitimat more then temporary acceptable. They must be enthusiastic about it as a particularly fine place in which to live and bring up their families. It must become the plan they want as homeland, the town they are going to make their own.

Under this model, workers are more acceptable of lower wages and higher housing costs to live in amenity rich areas, as currently experienced in different regions of the province. This change in economic development perspective is based on the assumptions and notions that what are attractive for skilled and knowledge based employees are beneficial for organizations, industry, and economic regions.

It is unclear whether the QOL strategy would be a more effective economic development strategy than traditional methods (Gottlieb, 1994). There is mounting evidence that suggests QOL factors are an important location determinant for both highly skilled workers and 'footloose' businesses. These businesses also present opportunities to Prince George in helping to attract further investment into the community, thus further diversifying the local economy. While it is unquestionable that QOL is important, is the subjective nature and vagueness of the concept presents challenges to identifying its effectiveness as an economic development policy strategy. The second challenge to this strategy is investment into amenities that present a much longer term approach to attracting firms, businesses and employees, whose payoffs may not be realized for years, making for potentially difficult decisions by local and municipal governments to support. In the concluding section, I examine these findings and offer translation as they apply to broader economic development policy and in the recruitment and retention strategy of personnel.

Recommendations

The first discussion of recommendations is in regards to the broad economic development policy related to QOL in Prince George. Wholesale abandonment of traditional economic development strategy in favor of an active QOL strategy is likely difficult to achieve or be supported by local government. It has been found that while most economic development officials recognize the importance of local amenities in attracting jobs, few are committed to undertaking a widespread amenity strategy at the expense of more traditional tools (Gottlieb, 1994). There is evidence that a wave of economic development strategy is emerging likely due to an awareness of the limitation of past policy efforts in light of the

shift from an industrial to a knowledge-based economy (Segedy, 1997). This shift in industries suggests a higher level of focus is required on QOL, and is in fact warranted for the local economy to prosper in this new economy. Economic development recommendations include:

- Coordination of QOL initiatives by Public--Private Sector for the improvement of local amenities and 'lifestyle' factors, such as parks, walkways, and city trails.
- 2) Development of a focused economic growth strategy on QOL factors, improvements of specific characteristics for short-term QOL improvements.
- Development of QOL economic-planning considerations that are necessary in business targeting and industry diversification over the long term.

Unmistakably, QOL is an important asset for retaining or recruiting top-line personnel, and is arguably an important issue in economic development policy (Glaser and Bardo, 1991).

The recommendations related to employer recruitment and retention is constrained by the fact that employers have limited influence of QOL factors and are limited by their resources available to use as incentives in personnel recruitment and retention.

- Utilizing a focused recruitment strategy, and coordination with the resources and initiatives of local economic development agency's marketing strategies.
- 2) Greater focus on the marketing and recruitment of university and college graduates in the region has not been adequately addressed in the past.
- Openness to immigration practices and need to develop working environments that are appealing to new employees and residents of the city and country.

There are important considerations in the strategy to recruit new college and university graduates that are related to the QOL factors and addressing the recruitment and retention challenges. Many community leaders would suggest that the group of people they would want to attract are likely successful married couples in their 30s and 40s - with middle to upper income jobs. It is important to consider the information that was presented by Micholas (1996) in the discussion of migration. It was also found that compared to the average citizen of any country, migrants are typically younger, better educated, employed in secondary industries or professions, unmarried or relatively recently married, with relatively fewer children, and are renters rather then home owners. This was supported by Way (2005) in her findings that only 33 percent of the newcomer respondents had children. In rapidly changing industries, the most recent graduates predictably possess the most up to date skills. Second, college-educated people postpone marriage longer than the national averages, and regions will have to offer a climate that satisfies this group's social interests and lifestyle needs, as well as address those of other groups to maintain a wider candidate pool (Florida, 2005). The City of Prince George and its industries play important roles in influencing the QOL issues, and addressing the issues of economic development and recruitment and retention in Prince George.

Limitations and Further Research

The nature of the study limited both the type of research completed and number of interviews completed. Further research would be suggested to increase the sample size and include more specific measures of QOL factors. Further research is also suggested to include interviews with employees of organizations to gain their perspectives on QOL

issues – where possible it would also be interesting to include prospective employees who declined to come to job offers in Prince George. It is also suggested to include more respondents from each firm to include participants of employees to provide their views and perspectives on the issues of employee recruitment and retention in the region. The interviews that were conducted resulted in a narrow focus on the recruitment and retention issues, and the nature of the study dictated limited information and research gathered related to the economic development aspects of QOL. There is a need to perform more empirical research into the relationship between QOL and business location decisions. Further study is needed since the interest in QOL as an economic development strategy has grown considerably, while the research done in large scale studies have been done to limited degrees. These studies and research can benefit the local community in helping to contribute to the local economy and quality of life enjoyed by all community members.

References

- Babbie, E. 2001, The Practice of Social Research. 9th edition. Belmont: Wadsworth' Thomas Learning.
- Bamberger, R.J., Parham, D. W. 1987. Leveraging amenity infrastructure: Indianapolis's economic development strategy. *Urban Land*, 43 (11): 12-18.
- Blair John, Premus Robert, 1987. Major Factors in Industrial Location: A Review. *Economic Development Quarterly* 1 (1): 72-85.
- Bogue, D. J. 1969, Principles of Demography. J. Wiley: New York.
- Boyle, M. R. Corporate Headquarters as Economic Development Targets. *Economic Development Review*. 6: 50-56.
- Brown, S. 1987. New directions for economic growth: Redesigning fiscal policies in Louisiana, New Mexico, and Texas. *Economic Review*, 13-20.
- Browning, E. J. 1980. How to select a business site: The executive's location guide. New York: McGraw-Hill.
- Catalytix and Richard Florida Creativity Group, 2003. Brain drain/gain. Creative Intelligence 1.6, 1-2.
- City of Prince George.2004 City Council Budget Meeting #7 Online Council Agenda http://www.city.pg.bc.ca/ cityhall/agendas/2004/2004_03_01_budget7.pdf (accessed March 12, 2007)
- Dissart and Deller 2000. Quality of Life in the Planning Literature. *Journal of Planning Literature*, 15 (1): 135–159.
- Festervand, T.A, Lumpkin, J.R., and Tosh, D.S. 1988. Quality of Life in the industrial Site Location Decision. *Journal of Real Estate Development 4*: 19-27
- Florida, R. May 2002. The Rise of the Creative Class. Washington Monthly. http://www.washingtonomonthly.com/features/2001/0205.florida.html (accessed February 14, 2007)
- Florida, R. 2002. The Rise of the creative class: and how it's transforming work, leisure, community, and everyday life. Basic Books, New York.
- and G. Gates 2005 *Technology and Tolerance*. In R. Florida, Cities and the creative class, Routledge, New York.
- Fuguitt, G. V. 1985, The nonmetropolitan population turnaround. *Annual review of Sociology* 11, 259-280.
- Glaser, M.A., and J.W. Bardo, 1991. The Impact of Quality of Life on the Recruitment and Retention of Key Personnel. *American Review of Public Administration*, 21: 57–72.

- Glasier, E. L. 2004, Review of Richard Florida's The Rise of the creative class. www.creativeclass. Org quoted in Jamie Peck, *Struggling with the Creative Class*. International Journal of Urban and Regional Research. 29.4, 740-770.
- Glasmeier, A.K. 1986. High-Tech Industries and the Regional Division of Labor. *Industrial Relations*. 25: 197-211
- Gottlieb, P. 1994. Amenities as Economic Development Tools: Is There Enough Evidence? Economic Development Quarterly, 8: 270–285.
- . 1995. Residential Amenities, Firm Location and Economic Development. *Urban Studies*, 32 (9): 1413–1436.
- Graves, P. E. 1980. Migration and Climate. The Journal of Regional Science, 20 (2), 227-237.
- Halseth, G. 1998. Prince George: A Social Geography of BC's Northern Capital. Prince George: UNBC Press.
- Halseth, G. and Ryser. L 2006. Trends In Service Delivery: Examples from rural and small town Canada, 1998 to 2005. *Journal of Rural and Community Development*, 1: 69-90
- Hack, G.D. 1984. The plan location decision making process. Industrial Development, 153, 31-33
- Harding, C.F. 1989.Location Choices for Research Labs: A Case Study Approach. Economic Development Quarterly 3:223-234
- Hayter, R. 1998. Corporate Strategies and Industrial Change in the Canadian Forest Products Industries. *Geographic Review*, 66, 1976: 209-228. Quoted in *Prince George: A Social Geography of BC's Northern Capital*, ed. G. and R. Halseth. Prince George: UNBC Press.
- Hedrick, Terry. Leonard Brickman and Debra Rog. 1993. Applied Research Design. Newbury Park; Sage Publications.
- Herzog, H. W. and Schlottan 1989. High Technology Location and Worker Mobility. *Knowledge and Industrial Organization*. Springer Verlag, Berlin: 47-61.
- Hoekstra, Gordon. 2004. Beetle Plague Hits Home. *Prince George Citizen* (Prince George, BC) July 31:13.
- Invest BC. Investment Highlights British Columbia http://www.investbc.com/highlights-ebay.htm (accessed March 12, 2007)
- Initiatives Prince George. 2006. "Prince George's Knowledge Economy, 2006." Prince George.
- Initiatives Prince George. 2005. 2005 Business Plan http://www.floatingrock.net/ipg/media/downloads/2005IPGBusPlanFinal.pdf (accessed March 12, 2007)

- Invest BC. Investment Highlights British Columbia http://www.investbc.com/highlights_pgcallcenter.htm (accessed March 12, 2007)
- Jamie Peck, 2005. Struggling with the Creative Class, *International Journal of Urban and Regional Research* December: 740-770
- Johnson, J. D. and R. Rasker 1995. The Role of Economic and Quality of Life Values in Rural Business Location. *Journal of Rural Studies* 11(4): 405–416.
- Long, L. H. 1973. Migration differentials by education and occupation: Trends and variations *Demography 10*, 243-258.
- Love, L., and J. Crompton 1999. The Role of Quality of Life in Business (re) Location Decisions. *Journal of Business Research*, 44: 211–222.
- Malanga, S. 2004. The curse of the creative class. City Journal Winter, 36-45/
- Marchuk, P. History of a Resource Industry. In A *History of British Columbia*, ed. P. Roy. Toronto: Copp Clark, 1989, 108-128. Quoted in *Prince George: A Social Geography of BC's Northern Capital*, ed. G. and R. Halseth. Prince George: UNBC Press, 1998.
- Michalos, A. 1997, Migration and The Quality of Life: A Review Essay. *Social Indicators Research*, 39: 121-166.
- Mullin, D.K. Changes in Location and Structure in the Forest Industry of North Central BC, 1909-1966. University of British Columbia, 1967, Unpublished MA Thesis. Quoted in *Prince George: A Social Geography of BC's Northern Capital*, ed. G. and R. Halseth. Prince George: UNBC Press, 1998.
- Morgan, W. 1964. The Effects of State and Local Tax and Financial Inducements on Industrial Location. (Ph.D. dis.), University of Colorado.
- Myers, D. 1987. Internal Monitoring of Quality of Life for Economic Development *Economic Development Quarterly*, 268-278
- Nielsen, Mark. 2007. Census finds fewer residents. *Prince George Citizen* (Prince George, BC), March 14:1.
- Paulson, Dave. 2003. Picture of Health? Prince George Citizen (Prince George): July 28:A4.
- Paulson, Dave. 2007. Sober Numbers. Prince George Citizen (Prince George): March14:1.
- Peck, J. 2005. Struggling with the Creative Class. *International Journal of Urban and Regional Research*, 29.4, 740-770.
- Premus, Robert. 1981 Location of High Technology Firms and Regional Economic Development, US Congress Joint Economic Committee. Washington: Joint Committee Print.

- Ritter, J. A. 1990. The Industrial Location Decision: A Practioner's Perspective. *Economic Development Quarterly*. 154-156
- Roger Schmenner. 1982. Making Business Location Decisions. Engelwood Cliffs, NJ: Prentice-Hall.
- Jan/feb. 126-32.
- Rooney, D. 1986. Strategies for local enterprise development. MIS Report, 18, 1-15.
- Salvesen D., and Renski. 2002. The Importance of Quality of Life in the Location Decisions of New Economy Firms. *Reviews of Economic Development*, 15.
- Schmenner, Roger, 1980. The Location Decision of Large Multiplant Companies. Washington: US Department of Housing and Urban Development.
- Statistics Canada. 1998 Labour Force Statistics. http://www.bcstats.gov.bc.ca/pubs/lfs/lfs9812.pdf (accessed March 12, 2007)
- Statistics Canada. 2006. Community Facts Prince George city. http://www.bcstats.gov.bc.ca/data/dd/ factsheet /cf163.pdf (accessed March 12, 2007)
- Statistics Canada, BC Stats. BC Unemployment Rates 2007. http://www.bcstats.gov.bc.ca/DATA/lss/lfs/ ur0702.pdf (accessed March 12, 2007)
- Stein, C.S., 1952. Planning Objectives. Reprinted in Kitimat Townsite 1962 Report. Kitimat, BC: District of Kitimat.
- Tiebout, C. M. 1956. A pure theory of local expenditures. *Journal of Political Economy* 66, 416-424.
- Watler Isard, 1975, Introduction to Regional Science. Englewood Cliffs, NJ: Prentice Hall, 200-204
- Way, Laura. 2005. Recruitment And Retention challenges of a Regional center of a Resource-Based Region: the Case Study of Prince George, BC. *University of Northern British Columbia*, 2005.
- Zuiches, J.J. 1980, Residential preferences in migration theory. in D. L. Brown and J. M. Wardell, *New Direction in Urban Rural Migration*. American Press, New York. 163-188.

Appendices

Appendix A Interview Questions for Key Informants

Appendix B Interview Consent Forms

Public Service Definition-Specific

Prince George public services specific to infrastructures and facilities related to public education, public health services and community recreation. (examples are; college and university, regional hospital and specialist services and parks, swimming pools and ice rinks.)

Quality of Life Definition-Broad

General quality of life (QoL)issues that include, public and social services, community pollution and crime, weather factors, cost of living and commuting, housing affordability, economic health and remoteness.

General Company Background

- 1. How long has your company been located in Prince George?
- 2. What were the most important factors in the decision to locate/expand in Prince George?
- 3. How does public services compare to other factors in your decision to invest in PG?
- 4. How does QoL compare to other factors in your decision to invest in PG?

General Aspects of Retention

- 5. Does your company have concerns in regards to employee retention?
- 6. What are methods used to retain your employees in the company and Prince George?
- 7. Are there common reasons your employees leave your company and Prince George?

General Aspects of Recruitment

- 8. Does your company have concerns in regards to employee recruitment?
- 9. What are methods used to recruit your employees in the company and Prince George? Is this process different for different levels of staff and management?
- 10. What is the cost to recruit and fill a position within your organization? Is this cost different for different levels of staff and management?

Public Service Issues

- 11. Do public services in Prince George affect your employee recruitment?
- 12. Do public services in Prince George affect your employee retention?
- 13. How important are public services (health/education/recreation) to your employee recruitment?
- 14. How important are public services (health/education/recreation) to your employee retention in Prince George?
- 15. How important are public services (health/education/recreation) to your company's operation in Prince George?

QoL Issues

- 16. Do QoL issues in Prince George affect your employee recruitment?
- 17. Do QoL issues in Prince George affect your employee retention?
- 18. How important are QoL issues to your employee recruitment?
- 19. How important are QoL issues to your employee retention in Prince George?
- 20. How important are QoL issues to your company's operation in Prince George?

Company Profile

- 21. What sector does the company operate in?
- 22. What ownership structure does the company have? (owner operator?)
- 23. What types of employees do you have? Permanent, part-time, contract
- 24. How many employees and managers do you have in your company?
- 25. Is your company more sensitive to QoL factors or public services in Prince George due to;
 - -high demand and competition for skilled labour
 - -strong economic condition of Prince George and BC
 - -technology changes to business location requirements
 - -industry changes to knowledge based workers
 - -other factors
- 26. Is there anything else around QoL factors or public services in Prince George that you think affects your company's operations?

What Impact Does Public Services In Prince George Have On Private Investment; And Effects On Recruitment And Retention

Interview Consent Form

<u>Purpose</u> – The purpose of the project is to examine the impact of public services in Prince George on economic development and recruitment and retention. The research will consist of key informant interviews with business managers and Human Resource managers of companies in Prince George. The goal is to provide businesses and community groups with information relevant to present and future community economic development planning and public service infrastructure investments.

How Respondents Were Chosen - The interview participants were selected from local business groups, economic development agencies, and local government officials in order to obtain views from people actively engaged in economic development and business operations. The interviewees were selected from publicly available lists as well as suggestions received from community economic development agencies in the Prince George area. The interviewees were selected for their potential to provide information about issues relevant to present and future public service needs related to education, health care and recreation so as to inform community planning.

Anonymity And Confidentiality - The names of participants will not be used in any reporting, nor will any information which may be used to identify individuals. All information shared in this interview will be held within strict confidence by the researcher. All records will be kept in a locked storage at the interviewer's place of residence and will be accessible only to the interviewer. The information will be kept until the final project report is complete. After this time, shredding will destroy all information related to the interview.

<u>Potential Risks And Benefits</u> - This project has been assessed by the UNBC Research Ethics Board. The researcher does not consider there to be any risks to participation. I hope that by participating you will have a chance to provide input into issues relevant to public service needs by businesses and employees in Prince George so as to guide future planning and investments.

<u>Voluntary Participation</u> - Your participation in the research project is entirely voluntary and, as such, you may chose not to participate. If you participate, you may choose to not answer any questions that make you uncomfortable, and you have the right to end the interview at any time and have all the information you provided withdrawn from the study.

<u>Research Results</u> - In case of any questions that may arise from this research, please feel free to contact Dr. Greg Halseth (250-960-5826) in the Geography Program at UNBC. The final project report will be distributed to all interviewees.

<u>Complaints</u> - Any complaints about this project should be directed to the Office of Research, UNBC (250) 960-5820 <reb@unbc.ca>.

I have read the above description of the study and I understand the conditions of my participation. My signature indicates that I agree to participate in this study.

(Name -please print)	(Signature)	(Date)